

Bower, H; Johnson, S; Bangura, MS; Kamara, AJ; Kamara, O; Mansaray, SH; Sesay, D; Turay, C; Checchi, F; Glynn, JR (2016) Effects of Mother's Illness and Breastfeeding on Risk of Ebola Virus Disease in a Cohort of Very Young Children. PLoS neglected tropical diseases, 10 (4). e0004622. ISSN 1935-2727 DOI: 10.1371/journal.pntd.0004622

Downloaded from: http://researchonline.lshtm.ac.uk/2536581/

DOI: 10.1371/journal.pntd.0004622

Usage Guidelines

 $Please \ refer \ to \ usage \ guidelines \ at \ http://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$

Available under license: http://creativecommons.org/licenses/by/2.5/

RESEARCH ARTICLE

Effects of Mother's Illness and Breastfeeding on Risk of Ebola Virus Disease in a Cohort of Very Young Children

Hilary Bower¹, Sembia Johnson², Mohamed S. Bangura², Alie Joshua Kamara², Osman Kamara², Saidu H. Mansaray², Daniel Sesay², Cecilia Turay², Francesco Checchi³, Judith R. Glynn¹*

1 Department of Infectious Disease Epidemiology, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Save the Children, Freetown, Sierra Leone, 3 Save the Children, London, United Kingdom

* judith.glynn@lshtm.ac.uk

Abstract

Background

Young children who contract Ebola Virus Disease (EVD) have a high case fatality rate, but their sources of infection and the role of breastfeeding are unclear.

Methods/Principal Findings

Household members of EVD survivors from the Kerry Town Ebola Treatment Centre in Sierra Leone were interviewed four to 10 months after discharge to establish exposure levels for all members of the household, whether or not they became ill, and including those who died. We analysed a cohort of children under three years to examine associations between maternal illness, survival and breastfeeding, and the child's outcome. Of 77 children aged zero to two years in the households we surveyed, 43% contracted EVD. 64 children and mothers could be linked: 25/40 (63%) of those whose mother had EVD developed EVD, compared to 2/24 (8%) whose mother did not have EVD, relative risk adjusted for age, sex and other exposures (aRR) 7·6, 95%Cl 2·0–29·1. Among those with mothers with EVD, the risk of EVD in the child was higher if the mother died (aRR 1·5, 0·99–2·4), but there was no increased risk associated with breast-feeding (aRR 0·75, 0·46–1·2). Excluding those breastfed by infected mothers, half (11/22) of the children with direct contact with EVD cases with wet symptoms (diarrhoea, vomiting or haemorrhage) remained well.

Conclusion/Significance

This is the largest study of mother-child pairs with EVD to date, and the first attempt at assessing excess risk from breastfeeding. For young children the key exposure associated with contracting EVD was mother's illness with EVD, with a higher risk if the mother died. Breast feeding did not confer any additional risk in this study but high risk from proximity to a



Citation: Bower H, Johnson S, Bangura MS, Kamara AJ, Kamara O, Mansaray SH, et al. (2016) Effects of Mother's Illness and Breastfeeding on Risk of Ebola Virus Disease in a Cohort of Very Young Children. PLoS Negl Trop Dis 10(4): e0004622. doi:10.1371/ journal.pntd.0004622

Editor: Oladele B. Akogun, Common Heritage Foundation, NIGERIA

Received: January 14, 2016

Accepted: March 20, 2016

Published: April 8, 2016

Copyright: © 2016 Bower et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files.

Funding: The study was funded by Save the Children (SCI) internal funds and The Wellcome Trust's Enhancing Research Activity in Epidemic Situations (ERAES) programme (Wellcome Trust Grant number: ER1502). JRG received this funding as a sub-grant holder of funding made to the London School of Hygiene and Tropical Medicine. HB was paid from this grant. SCI operated the Kerry Town Ebola Treatment Centre during the period under study, and employed the field team members. One author (FC), employed by Save the Children UK, was involved in commissioning the study and interpreting findings. Full responsibility for study design, data collection and analysis, decision to publish, & preparation of the manuscript was with JRG. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: Save the Children International (SCI) operated the Kerry Town ETC and employed the local field work team members (SJ, MSB, AJK, OK, SHM, DS, CT). One author (FC) is employed by SCI and was involved in commissioning the study and interpreting findings. No SCI or Wellcome Trust staff were involved in the analysis or the decision to submit for publication. The corresponding author (JRG) had full access to the study data and was responsible for the decision to submit for publication. HB and JRG declare that they have no conflicts of interests. sick mother supports WHO recommendations for separation. This study also found that many children did not become ill despite high exposures.

Author Summary

Our study is the first to quantify sources of infection and describe risk of transmission of Ebola to young children. We found that the risk of a child under three developing Ebola disease was low unless their mother had EVD, and that the risk was particularly high if their mother died of EVD. But we found no additional risk from breastfeeding. WHO recommends separating asymptomatic breast-fed infants from their mothers if they develop Ebola, and using formula feeding. We support the need for separation because of the high risk related to proximity, but more research is needed to more fully understand this, particularly given the importance of breast-feeding in preventing other childhood illnesses. We also found young children in Ebola-affected households whose mothers were <u>not</u> ill had a surprisingly low risk of developing EVD which was not all explained by low exposure to the virus. Many children stayed well despite having direct contact with EVD patients with diarrhoea, vomiting or bleeding who are considered the most infectious. We hope these findings will provide impetus for more detailed studies into age-related response to the Ebola virus.

Introduction

Young children experience a high case fatality rate from Ebola, but the incidence of Ebola Virus Disease (EVD) in children appears to be lower than in adults.[1-4] Young children may have limited exposure outside the home, but within the household maintaining hygiene in young children is difficult, although efforts may be made to keep children away from those who are sick. For very young children who need to be fed and held, contact with sick caregivers may be unavoidable.

Breastfeeding is a possible additional source of infection for young children: Ebola has been found in breast milk, but the risk to breastfed babies and the contribution of breastfeeding to transmission is poorly understood. [5,6] An investigation of household contacts following the Ebola outbreak in Gulu, Uganda in 2000 included five infants whose mother had EVD: three of four infants who were breastfed developed EVD. [7] The other infant was reported to have been separated from his mother early in the course of her illness and remained well; it is not clear if he was breastfed. Two recent systematic reviews of transmission of Ebola did not did not mention risks associated with breastfeeding. [8,9]

As part of a study of transmission patterns in Sierra Leone we collected data on exposure patterns and outcomes of all individuals present in the households of EVD survivors. In this analysis we sought to identify likely sources of infection and characterise risk of transmission to young children, including those breastfed by mothers with EVD.

Methods

In July-September 2015, interviews were sought with the household members of all individuals who were discharged from the Ebola Treatment Centre in Kerry Town, Sierra Leone ("Ebola survivors") from November 2014 to March 2015. Contact was made through members of the survivor support team who were involved in their reintegration into the community. An initial approach was made to explain the study. If the household head agreed, an interview was arranged at a community centre or other meeting place and all who were in the household at the time that members of the households had Ebola were encouraged to attend.

At the interview, individual informed written consent to participate in the study was sought from all adults, and from parents or guardians for children (< 18 years), with assent from children of 12 years or older. An inventory was drawn up of all household members who had been present in the household at the time that one or more household members were ill with EVD, including any who had died or were not present at the interview. For each member we asked whether they had had Ebola. We asked relatives whether any deceased had died of Ebola.

Household members were asked to describe what happened when Ebola came to their household, including who became ill first, whether those with Ebola had any diarrhoea, vomiting or bleeding while they were at home, and who looked after them. They were encouraged to tell the narrative in their own words, with probing questions to clarify who had been exposed and how. For each household member (including those who had died, but excluding any absent members or those who refused consent) we sought to establish the highest-risk exposure. Reported exposures were ranked a priori from highest to lowest as: contact with the body of someone who died of Ebola; direct contact with body fluids of someone with Ebola, including breastfeeding, or other direct contact with "wet" cases (i.e. those with diarrhoea, vomiting or bleeding); direct contact with "dry" cases (i.e. those without diarrhoea, vomiting or bleeding); indirect contact (e.g. shared utensils); and no known contact. For each mother-baby pair who both had EVD we attempted to ascertain from the narratives who was affected first.

All survivors from the Kerry Town Ebola Treatment Centre had EVD confirmed by PCR. We did not have laboratory data for those from other treatment centres or for those who died, so have relied on the families' reports. For individuals who were not reported as having had Ebola we asked about symptoms at the time that Ebola was in the household. For the analysis they were classified as not having had Ebola if they were asymptomatic or had symptoms that did not fulfil the Sierra Leone Ministry of Health and Sanitation case definition for "probable" Ebola,[10] or had had a negative test; and as having had Ebola if they were symptomatic and fulfilled the case definition for probable Ebola and were not tested. The case definition was contact with a case plus fever or miscarriage or unexplained bleeding; or contact plus three or more symptoms (of fatigue, headache, loss of appetite, nausea or vomiting, abdominal pain, diarrhoea, muscle or joint pain, sore throat or pain on swallowing, hiccups).

In this analysis we concentrate on risks to children aged less than three years at the time Ebola reached their household in order to include all those who were breast fed, and examine attack rates, case fatality rates and the role of breast feeding. Proportions were compared using X^2 or Fisher's exact test. Analyses used multivariable logistic regression. Because the outcome is very common we have presented the results as risk ratios (RR) using marginal standardization to estimate RRs, and the delta method to estimate 95% confidence intervals (95%CI).[11–13] We repeated the analysis calculating risk ratios using Poisson regression with robust error variance.[14] Crowding (number of people per room) and sanitation (access to water, soap and latrine) were considered as possible confounders, in addition to age, sex and the exposure variables. The effects of clustering by household were explored using generalised estimation equations in logistic regression: the results were very similar to analyses ignoring clustering so clustering is not included in the models. Analyses used STATA 14.

Ethics Statement

The study was approved by the Sierra Leone Ethics and Scientific Review Committee and the Ethics Committee of the London School of Hygiene & Tropical Medicine. At the interview, individual written informed consent to participate in the study was sought from all adults, and from parents or guardians for children (< 18 years), with assent from children of 12 years or older.

Results

One hundred and fifty one survivors were discharged from Kerry Town Ebola Treatment Centre from November 2014 through March 2015, of whom 138 were still living in the Western Area of Sierra Leone when sought for interview in July-September 2015. Twelve were uncontactable and a further two were known to have bad relationships with their households so were not approached. We contacted and interviewed 123 Kerry Town survivors, living in 94 households. Only one contacted survivor refused to be interviewed, and only two of 526 household members refused to participate. A further 37 members were not available to attend the interview. Some households also included survivors who had been treated in other facilities.

The households contained 77 children aged less than three years: 43% (33/77) got EVD, including four who fitted the case definition but were not diagnosed at the time. The risk of EVD was 54% (13/24) in those under one year; 40% (12/30) in those aged one year and 35% (8/23) in those aged two years (p-value for trend = 0.2). The risk was slightly higher in males than in females: 51.4% (18/35) vs 35.7% (15/42), p = 0.2. Three of the children were primary or co-primary cases in their household. Overall, 24 children under three years died of EVD, giving a case fatality rate of 73%: 85% (11/13), 75% (9/12) and 50% (4/8) at ages under-one, one, and two respectively (p-value for trend = 0.1).

Among the 77 children were 13 whose mothers were not present (including two mothers who had died in other households), or were not clearly identified: six (46%) of these children developed EVD and five died compared to 27 cases (42%) and 19 deaths among the 64 children who could be linked to their mothers.

Details of the mother-child pairs for whom the outcome of both mother and child are known are shown in <u>Table 1</u> for the 40 whose mothers had EVD, in <u>Table 2</u> for the 24 whose mothers had no symptoms, and in summary for all 64 in <u>Table 3</u>. The highest level of exposure is shown, in terms of direct or indirect exposure to those with EVD in the home or outside. None of the children had direct contact with dead bodies. Breastfeeding was taken as the highest exposure if the mother had EVD unless the child developed symptoms before or at the same time as the mother.

EVD in the children was much more likely among those whose mother had EVD (25/40, 63%) than among those whose mother did not get EVD (2/24, 8%, risk ratio (RR) 7.5, 95% confidence interval (CI) 1.9–28.9, p<0.0001, Table 3). The RR remained high after adjusting for age and sex of the child (RR 9.4, 95% CI 2.6–34.0), and after additionally adjusting for maximum exposure level (RR 7.6, 95% CI 2.0–29.1). Household crowding and sanitation were not associated with EVD in the child, and adjusting for them made little difference to the results. After adjusting for mother's EVD status and exposure levels, the risk of EVD in the child decreased with age (Table 3). After adjusting for mother's EVD, age, and sex, there was no effect of exposure level.

Among those whose mother had EVD, excluding the two pairs in which the children were ill first, the risk of EVD in the child was higher if the mother died (79% vs 50%, <u>Table 3</u>), giving a relative risk of 1.6 (95% CI 0.97-2.6). This association was similar after adjusting for the child's age and sex and additionally for exposure level. Of the 13 children who did not get EVD

Child's age in years	Mother's outcome	Child's exposure	Other Ebola cases	No. of people in household	Child's outcome	Timing
Breast fed						
<1	Survivor	breastfed	4	10	Well	mother first
<1	Survivor	breastfed	6	14	Survivor	mother first
<1	Survivor	breastfed	0	12	Survivor	unclear
<1	Survivor	breastfed	15	26	Death EVD	mother first
<1	Survivor	breastfed	2	6	Death EVD	mother first
<1	Survivor	breastfed	9	15	Death EVD	mother first
<1	Death EVD	breastfed	4	10	Death EVD ¹	mother first
<1	Death EVD	breastfed	11	17	Death EVD	mother first
<1	Death EVD	breastfed	18	26	Death EVD	unclear
<1	Death EVD	breastfed	10	13	Death EVD	mother first
1	Survivor	breastfed	3	15	Well	mother first
1	Survivor	breastfed	0	3	Well	mother first
1	Survivor	breastfed	1	9	Well	mother first
1	Survivor	breastfed	9	13	Death EVD	mother first
1	Death EVD	breastfed	3	11	Well	mother first
1	Death EVD	breastfed	13	27	Survivor	mother first
Breast fed, child ill before or same time as mother						
1	Survivor	indirect contact wet case	1	9	Death EVD	child first
2	Survivor	direct contact wet case	4	7	Death EVD	same time
Not breast fed						
<1	Death EVD	direct contact wet case	12	16	Death EVD	unclear
1	SymCD/ NoTest ²	direct contact body fluids	3	11	Well	unclear
1	Survivor	direct contact body fluids	4	13	Death EVD	unclear
1	Survivor	direct contact wet case	9	15	Death EVD	mother first
1	Survivor	direct contact wet case	4	10	Well	mother first
1	Survivor	direct contact wet case	4	9	SymCD/NoTest ³	mother first
1	Survivor	direct contact dry case	15	26	Death EVD	unclear

Table 1. Details of exposure and outcomes in mother-child pairs in which the mother had Ebola.

(Continued)

Child's age in years	Mother's outcome	Child's exposure	Other Ebola cases	No. of people in household	Child's outcome	Timing
1	Survivor	direct contact dry case	4	12	Death EVD	unclear
1	Death EVD	direct contact dry case	15	19	Headache/ cough only	unclear
1	Death EVD	direct contact dry case	15	26	Death EVD	unclear
2	SymCD/ NoTest ⁴	direct contact dry case	5	15	Well	unclear
2	Survivor	direct contact wet case	3	11	Well	unclear
2	Survivor	direct contact wet case	19	26	Headache only	unclear
2	Survivor	direct contact wet case	4	7	Survivor	mother first
2	Survivor	direct contact dry case	9	14	Well	unclear
2	Survivor	direct contact dry case	1	5	SymCD/NegTest ²	unclear
2	Survivor	indirect contact dry case	16	26	Well ⁵	unclear
2	Death EVD	direct contact body fluids	5	8	SymCD/NoTest ⁶	mother first
2	Death EVD	direct contact body fluids	5	8	SymCD/NoTest ⁶	mother first
2	Death EVD	direct contact wet case	3	6	Death EVD	mother first
2	Death EVD	minimal contact	4	19	Well ⁵	unclear
2	Death EVD	minimal contact	2	6	SymCD/NoTest ²	mother first

Table 1. (Continued)

Death EVD = death from Ebola; Survivor = Survived Ebola; SymCD/NoTest = fulfilled case definition for Ebola, not tested; SymCD/NegTest = fulfilled case definition for Ebola but tested negative

¹ Possibly infected in utero or perinatally.

- ² Fever only
- ³ Multiple symptoms, not tested because of nurses' strike
- ⁴ Fever and headache
- ⁵ Moved out of household after first case
- ⁶ Multiple symptoms

doi:10.1371/journal.pntd.0004622.t001

whose mother survived, five had contact with the mother when she was a wet case and five only when she was a dry case (unknown for three).

As the only child over two years who was breastfed got ill at the same time as the mother and was therefore excluded, the analysis of breastfeeding was restricted to the under two's. We also excluded the child who became ill first (<u>Table 1</u>), leaving 26 children. The proportion of children with EVD was very similar in those who were or were not breast fed (69% vs 70%, <u>Table 3</u>), RR 0.98, 0.58–1.7. There was no evidence of increased risk from breastfeeding after adjusting for age and sex (RR 0.76, 0.46–1.2) or for whether the mother died (<u>Table 3</u>).

The analyses were re-run excluding the six mother-child pairs for which either the mother or the child was classified as having EVD on the basis of symptoms (Table 1). The associations

Child's age (years)	Child's exposure	Other Ebola cases in household	No. of people in household	Child's outcome
<1	direct contact wet case	19	26	Death EVD
<1	direct contact wet case	14	15	Death EVD
<1	direct contact wet case	6	12	Well
<1	direct contact wet case	1	13	Well
<1	direct contact dry case	6	18	Well
<1	direct contact dry case	5	13	Well
<1	direct contact dry case	2	14	Well
<1	direct contact dry case	1	9	Well
<1	minimal contact	6	18	Well
<1	minimal contact	3	11	Well
<1	minimal contact	2	8	Well
1	direct contact wet case	6	12	Well
1	direct contact wet case	1	13	Well
1	direct contact dry case	2	11	Well
1	direct contact dry case	1	12	Well
1	direct contact dry case	1	3	Well
1	direct contact dry case	1	4	Headache only
1	minimal contact	2	7	Well
1	minimal contact	6	8	Well ¹
2	direct contact wet case	1	6	Well
2	direct contact wet case	1	13	Well
2	direct contact wet case	1	13	Well
2	direct contact dry case	2	11	Well
2	minimal contact	3	11	Well

Table 2. Details of ex	posures and outcomes in n	nother-child pairs in v	which the mother	did not have Ebola
Table El Detalle el ex				ala liet liate Ebela

Death EVD = death from Ebola

¹ Moved out of household after first case

doi:10.1371/journal.pntd.0004622.t002

with having a mother with EVD (fully adjusted RR 6.5, 1.6-26.0) and with breastfeeding (fully adjusted RR 0.74 (0.47-1.2) were similar to the main analysis, but the effect of having a mother who died of Ebola was lost (fully adjusted RR 1.3, 0.76-2.1). The analyses were also rerun using Poisson regression. The results were similar to the main analysis.

Among the children under three years whose mother did not get EVD, only two children got EVD. Both were aged under one year, from households with many EVD cases (<u>Table 2</u>), and both were reported to have had close contact with wet cases in the household. Seven other children whose mother did not have EVD and 4 whose mother had EVD but were not breastfed, had direct contact with wet cases and did not get ill. Overall, excluding children breastfed by mothers with EVD, half (11/22) of the children who had direct contact with wet cases or fluids remained well. These contacts included sharing beds with and embracing close relatives who suffered from vomiting and/or diarrhoea.

Discussion

Among the very young children in this study the risk of EVD depended largely on whether their mother developed EVD, with an additional risk for those whose mothers died of Ebola. The high risk in those with sick mothers is expected, and the higher risk in those with mothers who died may reflect higher viral loads and/or viral shedding in these mothers. The low risk in

Table 3. Associations with Ebola in children under three years.

	Child's outcome					
	Ebola /Total	%	RR (95% CI)	RR adjusted age and sex	Full model	P (full model)
All mother-baby pairs						
Mother had Ebola						
No	2/24	8.3	1	1	1	<0.001
Yes	25/40	62.5	7.5 (1.9–28.9)	9.4 (2.6–34.0)	7.6 (2.0–29.1) ¹	
Child's age						
< 1yr	12/22	54.6	1	1	1	0.03
1 yr	9/24	37.5	0.69 (0.36–1.3)	0.67 (0.35–1.3)	0.54 (0.33–0.88) ¹	
2 yrs	6/18	33.3	0.61 (0.29–1.3)	0.59 (0.28-1.3)	0.51 (0.27–0.96) ¹	
Child's sex						
Female	11/31	35.5	1	1	1	0.2
Male	16/33	48.5	1.4 (0.76–2.5)	1.4 (0.80–2.6)	1.4(0.87–2.2) ¹	
Child exposure level						
breastfeeding	11/16	68.8				
direct wet	11/22	50.0				
direct dry	3/16	18.8	0.68 (0.50–0.94) ²	0.70 (0.51–0.95)	0.93 (0.76–1.1) ¹	0.5
indirect wet	1/1	100				
indirect dry	0/1	0.0				
minimal	1/8	12.5				
Among those with mothers with Ebola ³						
All						
Mother survived	12/24	50.0	1	1	1	
Mother died	11/14	78.6	1.6 (0.97–2.6)	1.5 (0.98–2.3)	1.5 (0.99–2.4) ⁴	0.06
Under 2s						
Not breastfed	7/10	70.0	1	1	1	
Breastfed	11/16	68.8	0.98 (0.58–1.7)	0.76 (0.46-1.2)	0.75 (0.46–1.2) ⁵	0.3

¹ Model included age, sex, mother's Ebola, and exposure level

² Modelled as a linear term across categories

³ Excluding two in which the child was ill first/at the same time

⁴ Adjusted for age, sex, and exposure level

⁵ Adjusted for age, sex, and mother's death

doi:10.1371/journal.pntd.0004622.t003

children in Ebola-affected households when the mother was not ill is surprising, and cannot all be explained by low exposure in the children. Overall, nearly two thirds of under-three year olds had direct contact with wet cases in the household or their body fluids. While the risk of disease decreased with decreasing exposure, half of the young children with direct exposure to wet cases remained well.

Only three children were deliberately sent out of the household to reduce exposure, and for all three there was some exposure before they left. The opportunities for households to protect children from exposure are limited, particularly as more and more cases arise, and young children share beds with sick relatives. While a 'no touch policy' may be understood by older children, it is impossible to explain to an infant.

Among children whose mothers had EVD, being breastfed did not appear to increase the risk. Numbers were small and risks were already high in this group so there was limited power to detect an association. Current WHO guidelines recommend that asymptomatic breastfed

infants of Ebola-infected mothers should be separated from their mothers and replacement fed.[15] Although we found no excess risk from breastfeeding, further studies, ideally with larger, pooled datasets, are needed to assess this further before suggesting any changes to the recommendation. The high risk from proximity to a sick mother supports the need for separation.

The children in this study all came from households with at least one survivor. This may mean small households and households with fewer cases are underrepresented, as there would be a lower chance for small households to include a survivor, and households in which all cases of EVD died are missed. This might underestimate the case fatality rate and overestimate attack rates, but should not bias the relative risks by age and exposure.

This study shows the remarkable resilience of some young children despite apparent exposure to Ebola. This could be dose-related—we do not know the actual viral exposure through contact or breastfeeding—but in other contexts some people seem to be infected from minimal exposures. Relative resistance to Ebola could be influenced by genetic factors, [16] though the correlation between infections in mothers and children is more likely to reflect exposure patterns than shared genes. It is possible that there is some protection through maternal antibody from breastfeeding (perhaps more in mothers who survive) that counteracts any increased risk from transmission via breastmilk.

This is much the largest study of mother-child pairs with EVD to date, and the first attempt to assess any excess risk from breastfeeding. By visiting households after transmission had ceased and talking to all members we were able to determine exposure in much more detail than is possible in an acute epidemic situation. And because we included all children in these households, including those who were not sick, we have been able to calculate age and exposure-specific attack rates. In these households the risk to young children was largely dependent on whether their mother had EVD, regardless of whether they were breastfed.

Supporting Information

S1 File. STROBE checklist. Completed STROBE checklist. (DOCX)

Author Contributions

Conceived and designed the experiments: JRG HB FC. Performed the experiments: HB SJ MSB AJK OK SHM DS CT. Analyzed the data: JRG HB. Wrote the paper: HB JRG FC SJ MSB AJK OK SHM DS CT.

References

- WHO Ebola Response Team (2015) Ebola Virus Disease among Children in West Africa. N Engl J Med 372: 1274–1277.
- Glynn JR (2015) Age-specific incidence of Ebola virus disease. Lancet 386: 432. doi: <u>10.1016/S0140-6736(15)61446-5</u> PMID: <u>26251391</u>
- Helleringer S, Noymer A, Clark SJ, McCormick T (2015) Did Ebola relatively spare children? Lancet 386: 1442–1443.
- Rosello A, Mossoko M, Flasche S, Van Hoek AJ, Mbala P, et al. (2015) Ebola virus disease in the Democratic Republic of the Congo, 1976–2014. Elife 4.
- 5. Moreau M, Spencer C, Gozalbes JG, Colebunders R, Lefevre A, et al. (2015) Lactating mothers infected with Ebola virus: EBOV RT-PCR of blood only may be insufficient. Euro Surveill 20.
- Bausch DG, Towner JS, Dowell SF, Kaducu F, Lukwiya M, et al. (2007) Assessment of the risk of Ebola virus transmission from bodily fluids and fomites. J Infect Dis 196 Suppl 2: S142–147. PMID: <u>17940942</u>

- Francesconi P, Yoti Z, Declich S, Onek PA, Fabiani M, et al. (2003) Ebola hemorrhagic fever transmission and risk factors of contacts, Uganda. Emerg Infect Dis 9: 1430–1437. PMID: 14718087
- 8. Brainard J, Hooper L, Pond K, Edmunds K, Hunter PR (2015) Risk factors for transmission of Ebola or Marburg virus disease: a systematic review and meta-analysis. Int J Epidemiol.
- Van Kerkhove MD, Bento AI, Mills HL, Ferguson NM, Donnelly CA (2015) A review of epidemiological parameters from Ebola outbreaks to inform early public health decision-making. Sci Data 2: 150019. doi: 10.1038/sdata.2015.19 PMID: 26029377
- 10. World Health Organization, Sierra Leone Ministry of Health (2014) Clinical management of patients in the Ebola Treatment Centres and other care centres in Sierra Leone. Interim Emergency Guidelines. http://nerc.sl/?q=sierra-leone-ebola-treatment-centre-pocket-guide-15-dec-2014
- Localio AR, Margolis DJ, Berlin JA (2007) Relative risks and confidence intervals were easily computed indirectly from multivariable logistic regression. J Clin Epidemiol 60: 874–882. PMID: <u>17689803</u>
- Santos CA, Fiaccone RL, Oliveira NF, Cunha S, Barreto ML, et al. (2008) Estimating adjusted prevalence ratio in clustered cross-sectional epidemiological data. BMC Med Res Methodol 8: 80. doi: <u>10.</u> <u>1186/1471-2288-8-80</u> PMID: <u>19087281</u>
- Norton EC, Miller MM, Kleinman LC (2013) Computing adjusted risk ratios and risk differences in Stata. The Stata Journal 13: 492–509.
- Zou G (2004) A modified poisson regression approach to prospective studies with binary data. Am J Epidemiol 159: 702–706. PMID: <u>15033648</u>
- World Health Organization (2015) Nutritional care of children and adults with Ebola virus disease in treatment centres. <u>http://www.who.int/elena/titles/full_recommendations/nutrition_ebola/en/index4.</u> <u>html</u>. Geneva: World Health Organization.
- Sanchez A, Wagoner KE, Rollin PE (2007) Sequence-based human leukocyte antigen-B typing of patients infected with Ebola virus in Uganda in 2000: identification of alleles associated with fatal and nonfatal disease outcomes. J Infect Dis 196 Suppl 2: S329–336. PMID: <u>17940968</u>